

G OPEN ACCESS

Citation: Blackmon K, Waechter R, Landon B, Noël T, Macpherson C, Donald T, et al. (2020) Epilepsy surveillance in normocephalic children with and without prenatal Zika virus exposure. PLoS Negl Trop Dis 14(11): e0008874. <u>https://doi.org/</u> 10.1371/journal.pntd.0008874

Editor: Eugenia Corrales-Aguilar, University of Costa Rica, COSTA RICA

Received: April 28, 2020

Accepted: October 12, 2020

Published: November 30, 2020

Copyright: © 2020 Blackmon et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript.

Funding: Funding support was received from Eunice Kennedy Shriver National Institute of Child Health and Human Development R21HD093551-01 (RW, BL), United States Agency for International Development AID-OAA-A-14-00028 (ADL), Windward Islands Research and Education Foundation GSP-SRGI-18009 (KB), Stanford University Maternal Child Health Research Institute (ADL). The funders had no role in study design, **RESEARCH ARTICLE**

Epilepsy surveillance in normocephalic children with and without prenatal Zika virus exposure

Karen Blackmon^{1,2*}, Randall Waechter^{2,3}, Barbara Landon², Trevor Noël², Calum Macpherson², Tyhiesia Donald⁴, Nikita Cudjoe², Roberta Evans², Kemi S. Burgen², Piumi Jayatilake³, Vivian Oyegunle³, Otto Pedraza¹, Samah Abdel Baki⁵, Thomas Thesen^{6,7}, Dennis Dlugos⁸, Geetha Chari⁹, Archana A. Patel¹⁰, Elysse N. Grossi-Soyster¹¹, Amy R. Krystosik¹¹, A. Desiree LaBeaud¹¹

Mayo Clinic, Department of Psychiatry and Psychology, Jacksonville, Florida, United States of America,
Windward Islands Research and Education Foundation, St George's University, St George's, Grenada,
West Indies, 3 St George's University School of Medicine, Department of Physiology, Neuroscience, and
Behavioral Sciences, St. George's, Grenada, West Indies, 4 Ministry of Health, Government of Grenada,
West Indies, 5 Biosignal Group Inc., Boston, Massachusetts, United States of America, 6 New York
University School of Medicine, Department of Neurology, New York, New York, United States of America,
7 Department of Biomedical Sciences, University of Houston College of Medicine, USA, 8 Children's Hospital
of Philadelphia, Philadelphia, Pennsylvania, United States of America, 9 SUNY Downstate Health Sciences
University, Brooklyn, New York, United States of America, 10 Boston Children's Hospital, Department of
Neurology, Boston, Massachusetts, United States of America, 11 Stanford University School of Medicine,
Department of Pediatrics, Stanford, California, United States of America

* blackmon.karen@mayo.edu

Abstract

Children with Congenital Zika Syndrome and microcephaly are at high risk for epilepsy; however, the risk is unclear in normocephalic children with prenatal Zika virus (ZIKV) exposure [Exposed Children (EC)]. In this prospective cohort study, we performed epilepsy screening in normocephalic EC alongside a parallel group of normocephalic unexposed children [Unexposed Children (UC)]. We compared the incidence rate of epilepsy among EC and UC at one year of life to global incidence rates. Pregnant women were recruited from public health centers during the ZIKV outbreak in Grenada, West Indies and assessed for prior ZIKV infection using a plasmonic-gold platform that measures IgG antibodies in serum. Normocephalic children born to mothers with positive ZIKV results during pregnancy were classified as EC and those born to mothers with negative ZIKV results during and after pregnancy were classified as UC. Epilepsy screening procedures included a pediatric epilepsy screening questionnaire and video electroencephalography (vEEG). vEEG was collected using a multi-channel microEEG® system for a minimum of 20 minutes along with video recording of participant behavior time-locked to the EEG. vEEGs were interpreted independently by two pediatric epileptologists, who were blinded to ZIKV status, via telemedicine platform. Positive screening cases were referred to a local pediatrician for an epilepsy diagnostic evaluation. Epilepsy screens were positive in 2/71 EC (IR: 0.028; 95% CI: 0.003–0.098) and 0/71 UC. In both epilepsy-positive cases, questionnaire responses and interictal vEEGs were consistent with focal, rather than generalized, seizures. Both children met criteria for a clinical diagnosis of epilepsy and good seizure control was achieved with

data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: Dr. Samah G. Abdel Baki reports support from Bio-Signal Group Corporation and other grants outside the submitted work. In addition, Dr. Baki has a patent 61/554,743 pending and a U.S. patent 13/284,886. None of the other authors have competing interests to disclose.

carbamazepine. Our results indicate that epilepsy rates are modestly elevated in EC. Given our small sample size, results should be considered preliminary. They support the use of epilepsy screening procedures in larger epidemiological studies of children with congenital ZIKV exposure, even in the absence of microcephaly, and provide guidance for conducting epilepsy surveillance in resource limited settings.

Author summary

Epilepsy is a clinical concern in children with Congenital Zika Syndrome and microcephaly; however, it is unknown whether epilepsy rates are elevated in normocephalic children with prenatal Zika virus exposure. We found a modest elevation of epilepsy rates in the first year of life of exposed, relative to unexposed, children, despite no other neurological manifestations. Seizures in exposed children had features of focal, rather than generalized, epilepsy and were well controlled with carbamazepine. Our findings suggest that epilepsy surveillance should be included in the neurodevelopmental assessment of children with prenatal Zika virus exposure, even in the absence of neurologic manifestations at birth.

Introduction

Epilepsy is a clinical concern in children with Congenital Zika Syndrome (CZS). Epilepsy is diagnosed in 48–96% of children with CZS when microcephaly is present [1-5]. It is unclear whether the risk for epilepsy is elevated in normocephalic children with prenatal Zika virus (ZIKV) exposure [Exposed Children (EC)]. We aimed to address this question by conducting epilepsy surveillance in a prospective cohort of normocephalic EC alongside a parallel group of normocephalic unexposed children (UC).

The teratogenicity of ZIKV is well established [6]; ZIKV is able to cross the placental barrier, disseminate to the fetus, and target cortical progenitor [7] and glial cells [8], leading to malformations of cortical development (MCD). MCDs can be severe and visually obvious, such as microcephaly [9], but also focal and radiographically occult [10]. In animals and humans, ZIKV-induced brain malformations are found in the absence of microcephaly [10– 12] and are not reliably detected with standard antenatal diagnostic tools such as ultrasound [10]. This makes it difficult to discern whether EC were protected from ZIKV neurotropism during gestation or whether they harbor subtle pathology that might become epileptogenic at later stages of development.

In 2016, the tri-island nation of Grenada, Carriacou, and Petit Martinique experienced a ZIKV outbreak with peak transmission from May through October of 2016 [13]. In April 2016, the Ministry of Health and the Windward Islands Education and Research Foundation (WIND-REF) began recruiting pregnant women for participation in a ZIKV surveillance study during their antenatal and postnatal care appointments at local health centers, regardless of whether they presented with ZIKV symptoms. Outcome assessments were performed on their offspring between 1 and 12 months of age and again between 12 to 30 months of age to determine the spectrum of neurodevelopmental sequelae associated with prenatal ZIKV exposure. We administered an epilepsy surveillance protocol in the context of this larger, multidisciplinary effort.

Epilepsy surveillance in low- and middle-income countries (LMIC) is often limited by the absence of epilepsy specialists and diagnostic tools [14], although LMICs bear a disproportionate burden of epilepsy [15,16]. Insufficient surveillance can increase the time between epilepsy onset and treatment [17] and increased time to treatment can compound the negative effects of epilepsy [18]. A well-designed pediatric epilepsy screening questionnaire can improve epilepsy detection in large populations of children and in the absence of specialist care [19,20]. We utilized a pediatric epilepsy screening questionnaire that was designed to be administered by non-specialist health care providers in resource-limited settings and validated to discriminate focal from generalized seizures [20]. In a subset of children, we also obtained video electroencephalography (vEEG), which was independently interpreted remotely by at least two expert pediatric epileptologists, who were blinded to the child's ZIKV exposure status. This approach allowed us to bridge the gap between remote epilepsy specialists and the local pediatrician (TD) who performed diagnostic evaluations and provided clinical care for the children in our study. The aim of our study was to detect and characterize epilepsy in EC and to compare epilepsy incidence rates among EC and UC to global incidence rates at one year of life.

Methods

Ethics statement

Institutional Review Board approval was obtained at St George's University (IRB#16061) and Stanford University (IRB#45242). Written informed consent was obtained from all mothers who participated in this study. There was no financial compensation.

Participants

Mother-infant pairs were recruited between April 2016 and March 2017 from ten public health centers across all parishes of Grenada. A total of 384 mothers signed informed consent to participate (Fig_1). Mothers were recruited during the antenatal or postnatal period:

Antenatal Cohort: A total of 153 pregnant women were recruited during pregnancy. In this cohort, maternal serum was collected at a single time point during the prenatal period and a follow-up time point during the postnatal period (0–12 months postpartum).

Postnatal Cohort: A total of 231 women were recruited during the postnatal period. Maternal serum was collected at a single time point during the postnatal period (0–12 months postpartum).

Laboratory testing

Maternal serum samples were initially assessed for flavivirus exposure with indirect IgG capture Enzyme-Linked Immunosorbent Assay (ELISA) using dengue virus (DENV₁₋₄) antigen [21] andwere further assessed using a multiplexed assay on a nanostructured plasmonic gold (pGOLD) platform (Nirmidas Biotech, Palo Alto, CA) for the detection of IgG and IgG avidity against ZIKV and DENV antigens [22]. The pGOLD IgG immunoassay was used to cross-validate ELISA results and distinguish ZIKV from DENV antigens, as it has demonstrated sensitivity and specificity to ZIKV greater than 90% and 98%, respectively, in the convalescent phase [22]. IgG avidity testing provided further data on the timing of ZIKV and DENV infection [22]. In addition, reverse transcription polymerase chain reaction (RT-PCR) was used in seven mothers who were febrile at time of assessment [23]. Mothers were provided feedback on their ZIKV status by the Grenada Ministry of Health. Research staff administering followup assessments remained blinded to maternal ZIKV status.

Head circumference classification

Infants' occipitofrontal head circumference was measured at birth and at a follow-up visit between 12 and 30 months of age. Measurements were made by 3 independent raters and the



Fig 1. Flow diagram of enrollment, serum testing, and outcome evaluation.

https://doi.org/10.1371/journal.pntd.0008874.g001

mean of these measurements was utilized. Normocephalic was defined as occipitofrontal head circumference between the 3rd and 97th percentile for sex and age in accordance with World Health Organization Child Growth Standards [24].

Inclusion criteria

EC Group: Children were included in the EC group if: (1) they were born to mothers classified as "ZIKV-Infected" during pregnancy based on positive prenatal laboratory results (ELISA and pGOLD) for ZIKV, with avidity testing showing infection in the past 6 months [22]; and (2) they were normocephalic at birth and at the follow-up visit; and (3) they completed the epilepsy screening questionnaire. Given that inclusion in the EC group is based on positive *prenatal* maternal laboratory results for ZIKV, this excluded all children born to mothers in the postnatal cohort (i.e., maternal prenatal serum was not collected in this group).

UC Group: Children were included in the UC group if: (1) they were born to mothers who were classified as "ZIKV-Uninfected" during pregnancy, based on negative laboratory results for ZIKV on both the ELISA and pGOLD assays; and (2) they were normocephalic at birth and at the follow-up visit; and (3) they completed the epilepsy screening questionnaire. Maternal classification as "ZIKV-Uninfected" required consistently negative laboratory results for ZIKV on both the prenatal and postnatal serum samples in the antenatal cohort and the postnatal sample alone in the postnatal cohort.

Epilepsy surveillance protocol

An epilepsy surveillance protocol was administered to EC and UC during a follow-up evaluation when they were between 12 and 30 months of age (Fig 1). Our epilepsy surveillance protocol consisted of a pediatric epilepsy screening questionnaire [20], as well as vEEG in a subset of children whose caregivers consented to this procedure. We utilized a questionnaire that has successfully differentiated focal and generalized epilepsy in Zambian and Tanzanian cohorts [20]. One question was added to probe for infantile spasms. The questionnaire was reviewed for cultural appropriateness by a focus group of Grenadian women and suggested adaptations were implemented. It was administered to the primary guardian at the follow-up evaluation by local research staff with undergraduate or graduate degrees in Psychology who were trained in seizure safety, detection, and classification through online content made available through the Epilepsy Foundation. All research assistants were fluent in the local dialect and cultural idioms for describing seizures. Positive screenings were queried further to determine age of seizure onset and whether there was a family history of epilepsy and were referred for follow-up clinical evaluation with a local pediatrician.

The *micro*EEG® (www.biosignalgroup.com) monitoring system was used to collect vEEG because it has been validated in newborns [25] and was designed for use in field applications and high-noise environments [26–28]. The 26-channel system digitizes analogue EEG signals collected by electrodes, amplifies them, and using Bluetooth® wireless connectivity, transfers the physiological data to a dedicated laptop computer. The electrode caps come in different sizes and are made of elastic, weblike, fabric with electrodes attached in a configuration that conforms to the International 10/20 System. Research assistants were trained to apply the cap through instructional videos. After parting the child's hair under each electrode, a small amount of electro-gel was applied through a hole in the electrode center to minimize the electrode-scalp impedance. We recorded for a minimum of 20 minutes along with a video recording of participant behavior time-locked to the EEG. vEEG data were transferred to a HIPAA compliant server and were independently reviewed by at least two board-certified pediatric epileptologists (DD, GC, AP), who were blinded to the child's ZIKV status. The only

accompanying information was date of birth, gender, and medications at time of recording. EEG recordings were interpreted as normal or abnormal using a standardized form. This form included category-specific designations for normal and abnormal findings. If abnormal, the following specifiers were included: seizure (focal, generalized, both); interictal epileptiform activity (focal, generalized, both, or multifocal); status epilepticus (focal or generalized); slow-ing (focal, diffuse, both); triphasic waves; burst suppression; and sleep (normal, abnormal-asynchronous sleep architecture, and abnormal-other). The standardized form also included specifiers for whether a recording was technically limited due to artifacts (but interpretable) or uninterpretable. At the end of data collection, a live group consensus meeting was held to provide all available clinical information to the epileptologists and resolve discordant EEGs.

Epilepsy classification

Epilepsy classification followed International League Against Epilepsy (ILAE) guidelines for epilepsy diagnosis [29,30]. Epilepsy diagnosis was established by a local pediatrician (TD) after a full clinical evaluation in addition to consideration of data from our pediatric epilepsy screening questionnaire, vEEG interpretation forms, and information from the consensus meeting discussions.

Statistical analysis

To address the primary study question, we compared the proportion of epilepsy cases in EC and UC to global incidence rates at one year of age [31-37] using a Bayesian probabilistic approach, given that epilepsy onset in all positive cases was in the first year of life. Fleiss' Multirater Kappa coefficient was used as a measure of inter-rater reliability for absolute agreement of EEGs as normal versus abnormal on blinded review and was calculated prior to the group consensus meeting.

Results

Sample characteristics

Maternal ELISA and pGOLD results were concordant for flavivirus antigens; all of the samples that were positive for anti-flavivirus IgG by ELISA were also positive for anti-DENV, anti-ZIKV, or both using pGOLD, indicating that the pGOLD assay did not result in any false negatives. In addition, all samples that were negative for anti-flavivirus IgG by ELISA were also negative by pGOLD, indicating that the pGOLD assay did not result in any false positives.

Of the 384 women that signed consent to participate, a total of 113 were classified as "ZIK-V-Infected" during pregnancy based on positive prenatal laboratory results for ZIKV and avidity testing showing infection within the past 6 months (Fig_1). Within this group, 96% also tested positive for DENV; however, avidity testing showed that DENV infection occurred more than 6 months prior to prenatal serum collection in 80%, indicating more remote infection. A total of 117 women were classified as "ZIKV-Uninfected" during pregnancy based on negative laboratory results on prenatal and postnatal serum (antenatal cohort) or postnatal serum alone (postnatal cohort). ZIKV status during pregnancy could not be determined for 149 women because (1) they had positive laboratory results for ZIKV on postnatal serum and there was no prenatal serum available (N = 144); or (2) they had negative ZIKV results on prenatal serology but positive ZIKV results on postnatal serology (N = 5).

A total of 114 children (112 singletons, 2 twins) were born to mothers classified as "ZIK-V-Infected" during pregnancy. One of these children passed away in infancy (a twin); another child was found to have microcephaly at birth; and 41 were lost to follow-up because their caregivers declined to return for a follow-up outcome assessment or they could not be reached by phone. This resulted in 71 children that met criteria for the EC group. A total of 120 children (114 singletons, 6 twins) were born to mothers who were classified as "ZIKV-Uninfected" during pregnancy. One of these children was found to have macrocephaly at birth and 48 were lost to follow-up because their caregivers declined to return for a follow-up outcome assessment or they could not be reached by phone. This resulted in a total of 71 children in the UC group. The cohort of 71 EC and 71 UC represented 62% of the original cohort of EC (71/114) and 59% of the original cohort of UC (71/120). We observed no differences in household socio-demographics between mothers who consented for epilepsy screening and those who were lost to follow-up (Table 1).

Socio-demographic, clinical, and anthropometric comparisons

We observed no differences between EC and UC in household socio-demographics, sex distribution, rates of premature births, or neonatal complication rates (<u>Table 2</u>). Neonatal complications included respiratory distress (4 EC, 3 UC), jaundice (9 EC, 9 UC), and infection (1 EC, 2 UC). None of the EC had craniofacial disproportions, arthrogryposis, or motor abnormalities, which indicates that, in addition to being normocephalic, there were no other apparent neurological manifestations at birth. In addition, we observed no group differences between EC and UC in head circumference at birth, or age, head circumference, height, and weight at time of epilepsy surveillance (<u>Table 2</u>).

Pediatric epilepsy screening questionnaire

Epilepsy screening questionnaires were administered to the caregivers of 71 EC and 71 UC. Two of 71 cases from the EC group screened positive versus 0/71 cases from the UC group. In both positive EC cases, the child's mother indicated a history of multiple seizures, not associated with a fever, that were accompanied by loss of awareness and post-ictal fatigue. There was no indication of features consistent with infantile spasms or developmental regression. In both cases, the mother indicated features consistent with focal seizures (e.g., shaking/twitching on one side, eye/head-turning to one side). In both cases, seizure onset was in the first year of life, there was no family history of epilepsy, and the mothers had positive laboratory results for ZIKV during pregnancy but were asymptomatic. These two children were referred to a local pediatrician (TD) for an epilepsy diagnostic evaluation.

Table 1.	Socio-demographic	characteristics	of families	lost to follow	w-up versus	followed.

	Lost to Follow-Up (n = 89)	Followed (n = 142)	Chi-square	p-value
Household Income (mont	hly Eastern Caribbean Dollars)	3.42	0.49	
Under 1,000	7 (8%)	11 (8%)		
1,001-2,000	17 (19%)	18 (13%)		
2,001-3,000	15 (17%)	29 (20%)		
Over 3,000	8 (9%)	21 (15%)		
Unknown/Refused	42 (47%)	63 (44%)		
Maternal Education		1.07	0.90	
Primary	10 (11%)	20 (14%)		
Secondary	48 (54%)	77 (54%)		
College Degree	8 (9%)	9 (6%)		
Graduate Degree	4 (5%)	8 (6%)		
Unknown/Refused	19 (21%)	28 (20%)		

https://doi.org/10.1371/journal.pntd.0008874.t001

Video electroencephalography

A total of 66 mothers signed consent for vEEG procedures (48 EC; 18 EC). vEEGs were interpretable in 62/66 cases (94%). Recordings were predominantly acquired while the child was awake (74%); however, 26% of the children fell asleep during the recording. The quality of the vEEGs was good in 49/66 cases (74%), technically limited but interpretable in 13/66 cases (20%), and uninterpretable in 4/66 cases (6%). Inter-rater reliability on initial blinded review was moderate (Fleiss' Multirater Kappa: 0.47, 95% CI = 0.46–0.48) but comparable to other studies that used blinded review of EEGs [38]. Within the UC group, one vEEG was read as abnormal by two reviewers; both reviewers identified focal sharp waves in the centrotemporal region. Within the EC group, there were four vEEGs that were initially read as abnormal by one reviewer and normal by a second reviewer. Abnormal findings consisted of focal slowing in two cases and interictal focal epileptiform activity in two cases (one with epilepsy). For these four discordant cases, consensus was achieved during live unblinded joint review, with agreement that artifact could not be ruled out.

Epilepsy cases

The two cases that screened positive for epilepsy had a follow-up evaluation by a local pediatrician. In the first case, epilepsy had already been clinically diagnosed at two weeks of age based on a history of at least two unprovoked seizures more than 24 hours a part. There was no family history of epilepsy or birth complications. An MRI obtained at 10 months of age was unremarkable. Seizures were initially well controlled with carbamazepine; however, there was a cluster of breakthrough seizures at 24 months. Two vEEGs at 11 and 12 months of age were normal. The second case was newly diagnosed with epilepsy based on a history of at least two unprovoked seizures more than 24 hours a part. Seizure onset was at three months of age, based on maternal report. Carbamazepine was started with good seizure control documented at 24-month follow-up. There was no family history of epilepsy or birth complications. Neuroimaging was not available. vEEG at 17 months of age was read as abnormal by one reviewer on initial blinded review, with findings of rare focal epileptiform discharges in the frontocentral region; however, consensus review determined that artifact could not be definitively ruled out.

Epilepsy rates

Epilepsy was observed in 2/71 EC (IR: 2.8%; 95% CI: 0.34–9.81%) and 0/71 UC (IR: 0.00; 95% CI: 0.00–5.36%). Global epilepsy incidence rates range from 0.10% to 0.15% in the first year of life [31–37] (Fig 2). Using a Bayesian probabilistic approach, assuming a uniform prior probability of epilepsy of 0.000 to 0.002, a binomial likelihood of a positive diagnosis, and a prior relative risk ratio of 1.00, the predicted risk of epilepsy in EC is 0.16%, which is greater than what would be predicted by global incidence rates (Fig 2). The predicted risk of epilepsy in UC is 0.1%, which is within the range of global incidence rates. The posterior relative risk ratio in EC (0.0016/0.001) of 1.6 is a small effect, indicating that the risk of epilepsy in the first year of life is elevated in EC but still considered a rare condition.

Discussion

We performed epilepsy surveillance in a cohort of normocephalic children with prenatal ZIKV exposure (EC) alongside a parallel group of socio-demographically similar unexposed control children (UC). There was a modestly elevated incidence rate of epilepsy at one year of life in EC, in the absence of any other apparent neurological manifestations. This rate represents the first time point in a longitudinal series. Continued surveillance of this cohort will

	EC (N = 71)	UC (N = 71)	Chi-square	p-value	
Child's Sex	0.72	0.40			
Males	39 (55%)	44 (62%)			
Females	32 (45%)	27 (38%)			
Household Income (monthly; Eastern Caribbean Dollars)			1.74	0.78	
Under 1,000	6 (8%)	5 (7%)			
1,001–2,000	7 (10%)	11 (15%)			
2,001–3,000	13 (18%)	16 (23%)			
Over 3,000	11 (16%)	10 (14%)			
Unknown/Refused	34 (48%)	29 (41%)			
Maternal Education		2.64	0.62		
Primary	11 (15%)	11 (15%)			
Secondary	36 (51%)	36 (51%)			
College Degree	4 (9%)	4 (6%)			
Graduate Degree	6 (1%)	6 (8%)			
Unknown/Refused	14 (20%)	14 (20%)			
Food Security			2.80	0.42	
Food Secure	23 (32%)	22 (31%)			
Food Insecure (Moderate)	15 (21%)	23 (32%)			
Food Insecure (Severe)	7 (10%)	7 (10%)			
Unknown/Refused	26 (37%)	19 (27%)			
Prematurity					
> 37 weeks gestation	54 (76%)	57 (80%)	1.67	0.43	
\leq 37 weeks gestation	4 (6%)	6 (9%)			
Unknown/Refused	13 (18%)	8 (11%)			
Neonatal Complications					
No complication	52 (73%)	56 (79%)	2.82	0.25	
Neonatal complications	14 (20%)	14 (20%)			
Unknown/Refused	5 (7%)	1 (1%)			
	Mean (SD)	Mean (SD)	t-value	p-value	
Mother's age (years) at delivery	28.68 (6.15)	28.29 (7.04)	-0.34	0.73	
Child's age (months) at time of epilepsy screen	21.77 (4.08)	22.11 (4.44)	0.47	0.64	
Child's head circumference (cm) at birth	33.43 (1.34)	33.21 (1.77)	-0.73	0.47	
Child's head circumference (*z-score) at time of epilepsy screen	0.45 (0.89)	0.53 (0.91)	0.46	0.65	
Child's weight (*z-score) at time of epilepsy screen	0.36 (1.12)	0.31 (1.19)	-0.24	0.81	
Child's height (*z-score) at time of epilepsy screen	0.13 (1.02)	-0.04 (0.99)	-0.88	0.38	

Table 2. Socio-demographic comparisons between exposed and unexposed children.

EC = Normocephalic ZIKV-Exposed Children; UC = Unexposed Children

*z-scores calculated using World Health Organization Child Growth Standards

https://doi.org/10.1371/journal.pntd.0008874.t002

determine whether epilepsy rates increase with continuing brain maturation in EC, relative to rates in the parallel group of UC.

World Health Organization recommendations for screening, assessing, and managing neonates and infants in areas of ZIKV transmission include a full neurological examination, neuroimaging, and continued monitoring for seizures throughout infancy when microcephaly is present [24]. However, guidelines for continued surveillance of normocephalic children are less specified. It is recommended that normocephalic children with suspected, probable, or confirmed ZIKV exposure receive neurodevelopmental assessments at a minimum of 3, 9, and



Fig 2. Global incidence rates of epilepsy at one year of life. Epilepsy incidence rates at one year of age range from 0.10% to 0.15% [Nova Scotia: 0.12%³¹; United States (Minnesota): 0.10%³³; Iceland: 0.13%³⁵; Finland (Helsinki): 0.12%³⁴; Germany: 0.15%³⁷; Norway: 0.14%³⁶]. Bars represent 95% confidence interval (<u>https://www.sample-size.net/confidence-interval-proportion</u>). Using a Bayesian probabilistic approach, the predicted risk of epilepsy at one year of life in EC is 0.16%, which is higher than global incidence rates.

https://doi.org/10.1371/journal.pntd.0008874.g002

24 months of age [24]. Our results support the inclusion of a brief pediatric epilepsy screening questionnaire [20] in these follow-up evaluations to probe whether the child is having seizures and if so, to specify features of the episodes that can help distinguish focal from generalized epilepsy syndromes.

Prior case series that have investigated epilepsy profiles in children with Congenital Zika Syndrome (CZS) have only included children with microcephaly and/or craniofacial disproportions, arthrogryposis, and motor abnormalities [1-5]. The prevalence of epilepsy in these case series was high, ranging from 48–96%, and the most common seizure types were infantile spasms and focal seizures [4,5]. Seizure onset was typically in the first few months of life and the most predominant electroencephalographic (EEG) findings were abnormal background rhythms, hypsarrhythmia, and focal or multifocal epileptiform discharges [4,5]. Good seizure control was achieved in approximately 65% of children with CZS and microcephaly, more frequently with polytherapy [4].

Our study findings suggest that the epilepsy profile in normocephalic EC may have similar features to more severe CZS-associated epilepsy but also important differences. In our positive epilepsy cases, questionnaire responses indicated focal seizure features. There was no evidence for generalized epileptiform discharges or hypsarrhythmia on interictal vEEGs, which increased confidence in diagnosis of a focal epilepsy syndrome. Although the number of cases was small, our results cautiously raise the possibility that epilepsy in normocephalic EC might be associated with focal brain malformations that are on the milder end of the CZS spectrum [12,39]; however, high-resolution neuroimaging is needed for confirmation.

Clinical evidence for ZIKV-associated seizures in the context of more subtle congenital brain malformations is limited in humans; however, there is strong evidence in mice, with a high incidence of seizures and focal spiking activity in normocephalic ZIKV-infected mice that had neuropathological findings of calcifications and cytoarchitectural disruption [40]. Although we observed only a modest elevation of epilepsy rates in EC during their first year of life, it is possible that ZIKV-associated seizures can emerge at a later stage of development in humans. Seizures associated with focal cortical dysplasia typically emerge in early childhood [41]; however, onset can range from 3 to 35 years [42,43]. ZIKV mouse models also show increased susceptibility to chemically-induced seizures, raising the possibility that a "second hit" (e.g., illness, stress, or injury) may be needed to induce seizure onset in EC.

Continued surveillance is needed to better characterize lifetime epilepsy risk in EC, identify potential "second hits," and characterize the developmental trajectory of seizure expression. It is unclear whether seizures in EC will worsen over time, resist medication, or potentially resolve. ZIKV mouse models show resolution of "infantile" seizures in the majority of infected normocephalic mice [40], which suggests that EC-associated epilepsy may show a unique developmental pattern. Continued surveillance of our cohort provides a valuable opportunity to address these unanswered questions, with the current results serving as a baseline from which to compare future findings.

Our study also demonstrates that large-scale epilepsy surveillance is feasible in resourcelimited LMICs in the event of future ZIKV outbreaks. In several ZIKV-endemic countries throughout the Caribbean, there are no neurologists, EEG, or MRI services [14]. We trained non-specialist local research staff to administer the pediatric epilepsy screening questionnaire and vEEG, which allowed for broader surveillance and follow-up with specialists in the event of positive screens. The screening questionnaire was low cost (printing costs only), quick to administer (less than five minutes), and easily understood by caregivers after local adaptation of the language (i.e., "fits" for seizures). Importantly, one of the children diagnosed with epilepsy was identified by our screening procedures, which led to an evaluation, treatment, and good seizure control. This illustrates that epilepsy surveillance can reduce the time to treatment, potentially improving outcomes in at-risk children.

The *micro*EEG system that we utilized was easy to apply and well tolerated by the 12- to 30-month-old children that we screened. We did not use any form of sedation but were able to keep the majority of children from moving with behavioral methods, such as modeling of procedures (i.e., applying the EEG to a doll with the child in advance), encouraging the child to sit on their caregiver's lap, and reading or watching videos while recording. Ultimately, 94% of the vEEGs were considered interpretable, which demonstrates the feasibility of obtaining high quality vEEG data in remote, high-noise, resource-limited settings. There were no differences in the rates of abnormal vEEG findings between the EC and UC groups. This suggests that vEEG may not be necessary as a screening tool in EC without any apparent neurologic manifestations. However, results from our study support the use of vEEG in positive epilepsy screening cases to assist in epilepsy diagnosis.

There are several limitations in our study that warrant caution when generalizing findings to the larger population of EC. First, our study sample was small, and the incidence of epilepsy was rare. Reasons for the small sample size include challenges in determining ZIKV-exposure status in more than a third of our sample whose mothers were enrolled in the postnatal period, as well as challenges in getting families to return for follow-up epilepsy screening procedures. Results should be considered preliminary; they support the use of our epilepsy surveillance protocol in a large-scale follow-up study of EC from different regions of the Caribbean and Latin America to determine the prevalence of epilepsy, particularly as EC continue to mature. Second, by including all pregnant mothers, including those who were asymptomatic, we were

unable to identify the trimester of infection in the majority of the sample. Although this approach increased the uncertainty in timing of infection, it allowed for a broader sample of symptomatic and asymptomatic ZIKV-infected women and avoids the ascertainment bias of only including symptomatic mothers and/or children. Third, cross-reactivity within the flaviviral family is a concern when ZIKV infection status is determined by serological assay; however, the pGOLD platform that we utilized is specifically designed to minimize cross-reactivity in the differential diagnosis of ZIKV and DENV with high sensitivity and specificity [22]. Although results suggest that 20% of the mothers with positive laboratory results for ZIKV may have also been infected with DENV during pregnancy, this is unlikely to have impacted our findings as one of the children with epilepsy was born to a mother with evidence of more remote (i.e., more than 6 months) DENV infection; whereas, the other child was born to a mother with evidence of more proximal (i.e., less than 6 months) DENV infection. Fourth, a larger number of EC completed vEEG procedures relative to UC, which may have been due to a disinclination for mothers of UC to consent to the lengthier vEEG procedures. However, this had little impact on our main study findings, as epilepsy classification primarily relied on the presence of clinical seizures, not vEEG findings. Finally, a strength of our study is that head circumference measurements were obtained at birth and between 12 and 30 months, assuring that none of the EC developed late onset microcephaly, which has been previously described in ZIKV-exposed children who were normocephalic at birth [44].

Conclusion

Epilepsy surveillance procedures should be included in the neurodevelopmental assessment of children with suspected, probable, or confirmed *in utero* ZIKV exposure during their first two years of life, even if no neurologic manifestations are present at birth. A pediatric epilepsy screening questionnaire is low cost and high yield and can be administered by non-specialists in resource-limited ZIKV-endemic regions in the event of future ZIKV outbreaks.

Acknowledgments

The authors would like to extend their gratitude to the participants involved in this study and to the Grenada Ministry of Health for their support.

Author Contributions

Conceptualization: Karen Blackmon, Samah Abdel Baki, Thomas Thesen, Dennis Dlugos, Geetha Chari, Archana A. Patel.

Data curation: Karen Blackmon, Nikita Cudjoe, Roberta Evans, Amy R. Krystosik.

Formal analysis: Karen Blackmon, Otto Pedraza.

Funding acquisition: Karen Blackmon, Randall Waechter, Barbara Landon, A. Desiree LaBeaud.

Investigation: Karen Blackmon.

- Methodology: Karen Blackmon, Randall Waechter, Trevor Noël, Calum Macpherson, Tyhiesia Donald, Samah Abdel Baki, Thomas Thesen, Dennis Dlugos, Geetha Chari, Archana A. Patel, Elysse N. Grossi-Soyster, Amy R. Krystosik, A. Desiree LaBeaud.
- **Project administration:** Karen Blackmon, Randall Waechter, Barbara Landon, Trevor Noël, Calum Macpherson, Nikita Cudjoe, Roberta Evans, Kemi S. Burgen, A. Desiree LaBeaud.

Resources: Karen Blackmon.

Supervision: Karen Blackmon, Randall Waechter, Barbara Landon, Nikita Cudjoe, Roberta Evans.

Visualization: Karen Blackmon, Otto Pedraza.

- Writing original draft: Karen Blackmon.
- Writing review & editing: Karen Blackmon, Randall Waechter, Barbara Landon, Trevor Noël, Calum Macpherson, Tyhiesia Donald, Nikita Cudjoe, Roberta Evans, Kemi S. Burgen, Piumi Jayatilake, Vivian Oyegunle, Otto Pedraza, Samah Abdel Baki, Thomas Thesen, Dennis Dlugos, Geetha Chari, Archana A. Patel, Elysse N. Grossi-Soyster, Amy R. Krystosik, A. Desiree LaBeaud.

References

- Oliveira-Filho J, Felzemburgh R, Costa F, Nery N, Mattos A, Henriques DF, et al. Seizures as a complication of congenital Zika syndrome in early infancy. Am J Trop Med Hyg. 2018; 98(6):1860–1862. https://doi.org/10.4269/ajtmh.17-1020 PMID: 29692307
- da Silva AAM, Ganz JS, Sousa PD, Doriqui MJ, Ribeiro MR, Branco MD, et al. Early growth and neurologic outcomes of infants with probable congenital Zika virus syndrome. Emerg Infect Dis. 2016; 22 (11):1953–1956. https://doi.org/10.3201/eid2211.160956 PMID: 27767931
- Alves LV, Paredes CE, Silva GC, Mello JG, Alves JG. Neurodevelopment of 24 children born in Brazil with congenital Zika syndrome in 2015: A case series study. BMJ Open. 2018; 8(7):e021304. <u>https:// doi.org/10.1136/bmjopen-2017-021304</u> PMID: <u>30012787</u>
- van der Linden H, Carvalho MD, van der Linden V, Lacerda KM, Pessoa A, Carneiro ML. Epilepsy profile in infants with congenital zika virus infection. New England Journal of Medicine. 2018; 379(9):891– 892. https://doi.org/10.1056/NEJMc1716070 PMID: 30157396
- Pessoa A, van der Linden V, Yeargin-Allsopp M, Carvalho MDCG, Ribeiro EM, Van Naarden Braun K, et al. Motor abnormalities and epilepsy in infants and children with evidence of congenital zika virus infection. Pediatrics. 2018; 141(Suppl 2):S167–S179. <u>https://doi.org/10.1542/peds.2017-2038F</u> PMID: 29437050
- de Araújo TVB, Ximenes RAA, Miranda-Filho DB, Souza WV, Montarroyos UR, de Melo APL, et al. Association between microcephaly, Zika virus infection, and other risk factors in Brazil: Final report of a case-control study. Lancet Infect Dis. 2018; 18(3):328–336. <u>https://doi.org/10.1016/S1473-3099(17)</u> <u>30727-2</u> PMID: <u>29242091</u>
- Charlier C, Beaudoin MC, Couderc T, Lortholary O, Lecuit M. Arboviruses and pregnancy: maternal, fetal, and neonatal effects. Lancet Child Adolesc Health. 2017; 1(2):134–146. <u>https://doi.org/10.1016/ S2352-4642(17)30021-4</u> PMID: <u>30169203</u>
- Van Den Pol AN, Mao G, Yang Y, Ornaghi S, Davis JN. Zika virus targeting in the developing brain. J Neurosci. 2017; 37(8):2161–2175. <u>https://doi.org/10.1523/JNEUROSCI.3124-16.2017</u> PMID: 28123079
- de Fatima Vasco Aragao M, van der Linden V, Brainer-Lima AM, Coeli RR, Rocha MA, Sobral da Silva P, et al. Clinical features and neuroimaging (CT and MRI) findings in presumed Zika virus related congenital infection and microcephaly: retrospective case series study. BMJ. 2016; 353:i1901. <u>https://doi.org/10.1136/bmj.i1901</u> PMID: <u>27075009</u>
- Adams Waldorf KM, Nelson BR, Stencel-Baerenwald JE, Studholme C, Kapur RP, Armistead B, et al. Congenital Zika virus infection as a silent pathology with loss of neurogenic output in the fetal brain. Nat Med. 2018; 24(3):368–374. <u>https://doi.org/10.1038/nm.4485</u> PMID: <u>29400709</u>
- Soares de Oliveira-Szejnfeld P, Levine D, Melo AS, Amorim MM, Batista AG, Chimelli L, et al. Congenital Brain Abnormalities and Zika Virus: What the Radiologist Can Expect to See Prenatally and Postnatally. Radiology. 2016; 281(1):203–18. <u>https://doi.org/10.1148/radiol.2016161584</u> PMID: <u>27552432</u>
- Aragao MFVV, Holanda AC, Brainer-Lima AM, Petribu NCL, Castillo M, van der Linden V, et al. Nonmicrocephalic infants with congenital zika syndrome suspected only after neuroimaging evaluation compared with those with microcephaly at birth and postnatally: How large is the Zika virus 'iceberg'? AJNR Am J Neuroradiol. 2017; 38(7):1427–1434. https://doi.org/10.3174/ajnr.A5216 PMID: 28522665
- Brenciaglia M, Noël TP, Fields PJ, Bidaisee S, Myers TE, Nelson WM, et al. Clinical, Serological, and Molecular Observations from a Case Series Study during the Asian Lineage Zika Virus Outbreak in Grenada during 2016. Can J Infect Dis Med Microbiol. 2018; 2018:4635647. <u>https://doi.org/10.1155/2018/ 4635647</u> PMID: 29623138

- Krauss G, Sandy S, Corbin DO, Bird-Compton J, Jack F, Nelson B, et al. Epilepsy care in the southern Caribbean. Epilepsy Behav. 2015; 51:267–72. <u>https://doi.org/10.1016/j.yebeh.2015.07.011</u> PMID: 26312988
- Birbeck GL. Epilepsy Care in Developing Countries: Part I of II. Epilepsy Curr. 2010; 10(4):75–9. https://doi.org/10.1111/j.1535-7511.2010.01362.x PMID: 20697498
- Silberberg D, Anand NP, Michels K, Kalaria RN. Brain and other nervous system disorders across the lifespan—global challenges and opportunities. Nature. 2015; 527(7578):S151–4. <u>https://doi.org/10. 1038/nature16028</u> PMID: 26580320
- Dua T, De Boer HM, Prilipko LL, Saxena S. Epilepsy care in the world: Results of an ILAE/IBE/WHO Global Campaign Against Epilepsy survey. Epilepsia. 2006; 47(7):1225–31. <u>https://doi.org/10.1111/j. 1528-1167.2006.00595.x</u> PMID: <u>16886987</u>
- Sutula TP, Hagen J, Pitkänen A. Do epileptic seizures damage the brain? Curr Opin Neurol. 2003; 16 (2):189–95. https://doi.org/10.1097/01.wco.0000063770.15877.bc PMID: 12644748
- Douglass LM, Kuban K, Tarquinio D, Schraga L, Jonas R, Heeren T et al. A Novel Parent Questionnaire for the Detection of Seizures in Children. Pediatr Neurol. 2016; 54:64–69.e1. <u>https://doi.org/10.1016/j. pediatrneurol.2015.09.016</u> PMID: <u>26552646</u>
- Patel AA, Ciccone O, Njau A, Shanungu S, Grollnek AK, Fredrick F, et al. A pediatric epilepsy diagnostic tool for use in resource-limited settings: A pilot study. Epilepsy Behav. 2016; 59:57–61. <u>https://doi.org/</u> 10.1016/j.yebeh.2016.03.019 PMID: 27088519
- Grossi-Soyster EN, Cook EAJ, de Glanville WA, Thomas LF, Krystosik AR, Lee J, et al. Serological and spatial analysis of alphavirus and flavivirus prevalence and risk factors in a rural community in western Kenya. PLoS Negl Trop Dis. 2017; 11(10):e0005998. <u>https://doi.org/10.1371/journal.pntd.0005998</u> PMID: <u>29040262</u>
- 22. Zhang B, Pinsky BA, Ananta JS, Zhao S, Arulkumar S, Wan H, et al. Diagnosis of Zika virus infection on a nanotechnology platform. Nat Med. 2017; 23(5):548–550. <u>https://doi.org/10.1038/nm.4302</u> PMID: 28263312
- Waggoner JJ, Abeynayake J, Sahoo MK, Gresh L, Tellez Y, Gonzalez K, et al. Single-Reaction, Multiplex, Real-Time RT-PCR for the Detection, Quantitation, and Serotyping of Dengue Viruses. PLoS Negl Trop Dis. 2013; 7(4):e2116. <u>https://doi.org/10.1371/journal.pntd.0002116</u> PMID: <u>23638191</u>
- 24. World Health Organization. Screening, assessment and management of neonates and infants with complications associated with Zika virus exposure in utero. WHO 2016. <u>http://apps.who.int/iris/bitstream/10665/204475/1/WHO_ZIKV_MOC_16.3_eng.pdf?ua=1</u>
- Ibrahim ZH, Chari G, Abdel Baki S, Bronshtein V, Kim MR, Weedon J, et al. Wireless multichannel electroencephalography in the newborn. J Neonatal Perinatal Med. 2016; 9(4):341–348. <u>https://doi.org/ 10.3233/NPM-161643</u> PMID: 28009337
- Grant AC, Abdel-Baki SG, Omurtag A, Sinert R, Chari G, Malhotra S, et al. Diagnostic accuracy of microEEG: A miniature, wireless EEG device. Epilepsy Behav. 2014; 34:81–5. <u>https://doi.org/10.1016/j. yebeh.2014.03.015 PMID: 24727466</u>
- Omurtag A, Baki SG, Chari G, Cracco RQ, Zehtabchi S, Fenton AA, et al. Technical and clinical analysis of microEEG: A miniature wireless EEG device designed to record high-quality EEG in the emergency department. Int J Emerg Med. 2012; 5:35. <u>https://doi.org/10.1186/1865-1380-5-35</u> PMID: 23006616
- Birbeck GL, Herman ST, Capparelli EV, Dzinjalamala FK, Abdel Baki SG, Mallewa M, et al. A clinical trial of enteral Levetiracetam for acute seizures in pediatric cerebral malaria. BMJ Paediatr. 2019; 19 (1):399 <u>https://doi.org/10.1186/s12887-019-1766-2</u> PMID: <u>31672143</u>
- 29. Thurman D J, Beghi E, Begley CE, Berg AT, Buchhalter JR, Ding D, et al. Standards for epidemiologic studies and surveillance of epilepsy. Epilepsia. 2011; 52(Suppl 7):2–26. <u>https://doi.org/10.1111/j.1528-1167.2011.03121.x</u> PMID: 21899536
- Fisher RS, Acevedo C, Arzimanoglou A, Bogacz A, Cross JH, Elger CE, et al. ILAE Official Report: A practical clinical definition of epilepsy. Epilepsia. 2014; 55(4):475–82. <u>https://doi.org/10.1111/epi.12550</u> PMID: 24730690
- Camfield P, Camfield C. Incidence, prevalence and aetiology of seizures and epilepsy in children. Epileptic Disord. 2015; 17(2):117–23. <u>https://doi.org/10.1684/epd.2015.0736</u> PMID: <u>25895502</u>
- Hauser WA, Annegers JF, Kurland LT. Incidence of Epilepsy and Unprovoked Seizures in Rochester, Minnesota: 1935–1984. Epilepsia. 1993; 34(3):453–68. <u>https://doi.org/10.1111/j.1528-1157.1993.</u> tb02586.x PMID: 8504780
- Wirrell EC, Grossardt BR, Wong-Kisiel LC, Nickels KC. Incidence and classification of new-onset epilepsy and epilepsy syndromes in children in Olmsted County, Minnesota from 1980 to 2004: A population-based study. Epilepsy Res. 2011; 95(1–2):110–8. <u>https://doi.org/10.1016/j.eplepsyres.2011.03</u>. 009 PMID: 21482075

- 34. Gaily E, Lommi M, Lapatto R, Lehesjoki AE. Incidence and outcome of epilepsy syndromes with onset in the first year of life: A retrospective population-based study. Epilepsia. 2016; 57(10):1594–1601. https://doi.org/10.1111/epi.13514 PMID: 27574005
- Olafsson E. Ludvigsson P, Gudmundsson G, Hesdorffer D, Kjartansson O, Hauser WA. Incidence of unprovoked seizures and epilepsy in Iceland and assessment of the epilepsy syndrome classification: A prospective study. Lancet Neurol. 2005; 4(10):627–34. <u>https://doi.org/10.1016/S1474-4422(05)70172-</u> 1 PMID: 16168931
- Aaberg KM, Gunnes N, Bakken IJ, Lund Søraas C, Berntsen A, Magnus P, et al. Incidence and prevalence of childhood epilepsy: A nationwide cohort study. Pediatrics. 2017; 139(5):e20163908. <u>https://doi.org/10.1542/peds.2016-3908</u> PMID: <u>28557750</u>
- Freitag CM, May TW, Pfäfflin M, König S, Rating D. Incidence of epilepsies and epileptic syndromes in children and adolescents: A population-based prospective study in Germany. Epilepsia 2001; 42 (8):979–85. https://doi.org/10.1046/j.1528-1157.2001.042008979.x PMID: 11554882
- Grant AC, Abdel-Baki SG, Weedon J, Arnedo V, Chari G, Koziorynska E, et al. EEG interpretation reliability and interpreter confidence: A large single-center study. Epilepsy Behav. 2014; 32:102–7. <u>https:// doi.org/10.1016/j.yebeh.2014.01.011</u> PMID: 24531133
- **39.** Petribu NCL, Aragao MFV, van der Linden V, Parizel P, Jungmann P, Araújo L, et al. Follow-up brain imaging of 37 children with congenital Zika syndrome: Case series study. BMJ. 2017; 359:j4188. https://doi.org/10.1136/bmj.j4188 PMID: 29030384
- 40. Nem de Oliveira Souza I, Frost PS, França JV, Nascimento-Viana JB, Neris RLS, Freitas L, et al. Acute and chronic neurological consequences of early-life zika virus infection in mice. Sci Transl Med. 2018; 10(444):eaar2749. https://doi.org/10.1126/scitranslmed.aar2749 PMID: 29875203
- Bast T, Ramantani G, Seitz A, Rating D. Focal cortical dysplasia: Prevalence, clinical presentation and epilepsy in children and adults. Acta Neurol Scand. 2006; 113(2):72–81. <u>https://doi.org/10.1111/j.1600-0404.2005.00555.x</u> PMID: <u>16411966</u>
- Blackmon K, Kuzniecky R, Barr WB, Snuderl M, Doyle W, Devinsky O, et al. Cortical gray-white matter blurring and cognitive morbidity in focal cortical dysplasia. Cereb Cortex. 2015; 25(9):2854–2862. <u>https://doi.org/10.1093/cercor/bhu080</u> PMID: <u>24770710</u>
- **43.** Blackmon K, Barr WB, Morrison C, MacAllister W, Kruse M, Pressl C, et al. Cortical gray–white matter blurring and declarative memory impairment in MRI-negative temporal lobe epilepsy. Epilepsy Behav. 2019; 97:34–43. <u>https://doi.org/10.1016/j.yebeh.2019.05.009</u> PMID: <u>31181427</u>
- 44. van der Linden V, Pessoa A, Dobyns W, et al. Description of 13 Infants Born During October 2015-January 2016 With Congenital Zika Virus Infection Without Microcephaly at Birth—Brazil. MMWR Morb Mortal Wkly Rep. 2016; 65(47):1343–1348. <u>https://doi.org/10.15585/mmwr.mm6547e2</u> PMID: <u>27906905</u>